Author's response to reviews

Title: Barriers to obesity management: a pilot study of primary care clinicians

Authors:

Valerie F Hoffman (valerie-hoffman@uiowa.edu)
Amanda Little (amanda-little@uiowa.edu)
Terry Wahls (terry.wahls@va.gov)

Version: 4 Date: 5 June 2006

Author's response to reviews: see over
June 5, 2006

Melissa Norton, MD, Medical Editor  
*BMC-Family Practice*

RE: Third Revision of *Barriers to Obesity Management: A Pilot Study of Primary Care Clinicians*

Dear Dr. Norton,

Enclosed is the third revision of our manuscript, which we submit for your consideration for publication in *BMC-Family Practice*.

We have made the requested stylistic changes.

Thank you again for the opportunity to submit and revise our manuscript with you.

Sincerely,

Valerie Forman Hoffman, PhD, MPH